Bond University Research Repository



Blood pressure-lowering treatment strategies based on cardiovascular risk versus blood pressure: A meta-analysis of individual participant data

Blood Pressure Lowering Treatment Trialists' Collaboration

Published in: PLoS Medicine

DOI: 10.1371/journal.pmed.1002538

Licence: CC BY

Link to output in Bond University research repository.

Recommended citation(APA):

Blood Pressure Lowering Treatment Trialists' Collaboration (2018). Blood pressure-lowering treatment strategies based on cardiovascular risk versus blood pressure: A meta-analysis of individual participant data. *PLoS Medicine*, *15*(3), Article e1002538. https://doi.org/10.1371/journal.pmed.1002538

General rights

Copyright and moral rights for the publications made accessible in the public portal are retained by the authors and/or other copyright owners and it is a condition of accessing publications that users recognise and abide by the legal requirements associated with these rights.

For more information, or if you believe that this document breaches copyright, please contact the Bond University research repository coordinator.



Citation: Karmali KN, Lloyd-Jones DM, van der Leeuw J, Goff DC, Jr., Yusuf S, Zanchetti A, et al. (2018) Blood pressure-lowering treatment strategies based on cardiovascular risk versus blood pressure: A meta-analysis of individual participant data. PLoS Med 15(3): e1002538. https://doi.org/10.1371/journal.pmed.1002538

Academic Editor: Steven R. Steinhubl, Scripps Translational Science Institute, UNITED STATES

Received: October 3, 2017

Accepted: February 16, 2018

Published: March 20, 2018

Copyright: This is an open access article, free of all copyright, and may be freely reproduced, distributed, transmitted, modified, built upon, or otherwise used by anyone for any lawful purpose. The work is made available under the <u>Creative</u> Commons CCO public domain dedication.

Data Availability Statement: Data from

contributing trials have been shared with the BPLTT Collaboration on the understanding that such data will stored and handled confidentially and that they will be used for the purpose of BPLTTC meta-analyses only. The data sharing agreement between data custodians and BPLTTC does not permit onwards transfer of the data to third parties. Any requests for trial data should therefore be made directly to data custodians. External researchers may suggest new research RESEARCH ARTICLE

Blood pressure-lowering treatment strategies based on cardiovascular risk versus blood pressure: A meta-analysis of individual participant data

Kunal N. Karmali¹, Donald M. Lloyd-Jones¹, Joep van der Leeuw², David C. Goff, Jr.³, Salim Yusuf⁴, Alberto Zanchetti⁵, Paul Glasziou⁶, Rodney Jackson⁷, Mark Woodward^{8,9}, Anthony Rodgers⁹, Bruce C. Neal⁹, Eivind Berge¹⁰, Koon Teo⁴, Barry R. Davis¹¹, John Chalmers⁹, Carl Pepine¹², Kazem Rahimi⁸*, Johan Sundström¹³, on behalf of the Blood Pressure Lowering Treatment Trialists' Collaboration¹¹

 Department of Medicine, Northwestern University, Chicago, Illinois, United States of America, 2 University Medical Center Utrecht, Department of Vascular Medicine, Utrecht, Netherlands, 3 National Heart, Lung, and Blood Institute, Division of Cardiovascular Sciences, Bethesda, Maryland, United States of America,
Population Health Research Institute, McMaster University, Hamilton, Canada, 5 The Istituto Auxologico Italiano, Milan, Italy, 6 Centre for Research in Evidence Based Practice, Bond University, Robina, Australia,
School of Population Health, Faculty of Medical and Health Science, University of Auckland, Auckland, New Zealand, 8 The George Institute for Global Health, University of Oxford, Oxford, United Kingdom, 9 The George Institute for Global Health, Sydney, Australia, 10 Department of Cardiology, Oslo University Hospital, Oslo, Norway, 11 School of Public Health, University of Texas, Dallas, Texas, United States of America,
Division of Cardiovascular Medicine, University of Florida, Gainesville, Florida, United States of America,
Department of Medical Sciences, Uppsala University, and Uppsala Clinical Research Center, Uppsala, Sweden

 \P Membership of the Blood Pressure Lowering Treatment Trialists' Collaboration is provided in the Acknowledgments.

* kazem.rahimi@georgeinstitute.ox.ac.uk

Abstract

Background

Clinical practice guidelines have traditionally recommended blood pressure treatment based primarily on blood pressure thresholds. In contrast, using predicted cardiovascular risk has been advocated as a more effective strategy to guide treatment decisions for cardiovascular disease (CVD) prevention. We aimed to compare outcomes from a blood pressure-lowering treatment strategy based on predicted cardiovascular risk with one based on systolic blood pressure (SBP) level.

Methods and findings

We used individual participant data from the Blood Pressure Lowering Treatment Trialists' Collaboration (BPLTTC) from 1995 to 2013. Trials randomly assigned participants to either blood pressure-lowering drugs versus placebo or more intensive versus less intensive blood pressure-lowering regimens. We estimated 5-y risk of CVD events using a multivariable Weibull model previously developed in this dataset. We compared the two strategies at specific SBP thresholds and across the spectrum of risk and blood pressure levels studied in proposals to the BPLTT Management Group. Such proposals shall be checked for feasibility and overlap with any existing proposals. If appropriate, they will be shared with the Steering Committee for scientific review. After approval, proposing researchers will be able to join a working group, with interested members of the Collaboration. This working group will work with the BPLTT Management group and Steering Committee to write the research protocol, conduct and interpret the results, and publish the final report. Any questions about the data used in this study or any interest in future collaborations shall be made to the BPLTTC Management (see bplttc.org).

Funding: This work was completed while DCG was at the Colorado School of Public Health. AR is supported by an NHMRC Principal Research Fellowship. The work of the George Institute Oxford is supported by the Oxford Martin School, University of Oxford and the NIHR Oxford Biomedical Research Centre. No funders or sponsors were involved in the design and conduct of the study; collection, management, analysis, and interpretation of the data; preparation, review, or approval of the manuscript; or decision to submit the manuscript for publication. All authors had full access to all data in the study and the first author and corresponding author had final responsibility for the decision to submit for publication.

Competing interests: I have read the journal's policy and the authors of this manuscript have the following competing interests: KR receives a stipend as a specialty consulting editor for *PLOS* Medicine and serves on the journal's editorial board. KR served as Guest Editor on PLOS Medicine's Special Issue on Cardiovascular Disease and Multimorbidity. MW is a consultant to Amgen on analyses of US health data. George Health Enterprises, the social enterprise arm of The George Institute for Global Health, has received investment to develop fixed-dose combination products containing aspirin, statin and blood pressure lowering drugs. George Health Enterprises has submitted patents for low-dose blood pressure combinations, on which AR is listed as one of the inventors. AR does not have a financial interest in these planned products. JC has received research grants, administered through the University of Sydney, as Principal Investigator for the ADVANCE trial and ADVANCE-ON post trial study, from the National Health and Medical Research Council of Australia and from Servier and JC has also received personal fees from Servier for speaking about these studies at scientific meetings.

Abbreviations: ABCD, Appropriate Blood Pressure Control in Diabetics; ACC, American College of

BPLTTC trials. The primary outcome was number of CVD events avoided per persons treated. We included data from 11 trials (47,872 participants). During a median of 4.0 y of follow-up, 3,566 participants (7.5%) experienced a major cardiovascular event. Areas under the curve comparing the two treatment strategies throughout the range of possible thresholds for CVD risk and SBP demonstrated that, on average, a greater number of CVD events would be avoided for a given number of persons treated with the CVD risk strategy compared with the SBP strategy (area under the curve 0.71 [95% confidence interval (CI) 0.70-0.72] for the CVD risk strategy versus 0.54 [95% CI 0.53–0.55] for the SBP strategy). Compared with treating everyone with SBP \geq 150 mmHg, a CVD risk strategy would require treatment of 29% (95% CI 26%-31%) fewer persons to prevent the same number of events or would prevent 16% (95% CI 14%-18%) more events for the same number of persons treated. Compared with treating everyone with SBP \geq 140 mmHg, a CVD risk strategy would require treatment of 3.8% (95% Cl 12.5% fewer to 7.2% more) fewer persons to prevent the same number of events or would prevent 3.1% (95% Cl 1.5%-5.0%) more events for the same number of persons treated, although the former estimate was not statistically significant. In subgroup analyses, the CVD risk strategy did not appear to be more beneficial than the SBP strategy in patients with diabetes mellitus or established CVD.

Conclusions

A blood pressure-lowering treatment strategy based on predicted cardiovascular risk is more effective than one based on blood pressure levels alone across a range of thresholds. These results support using cardiovascular risk assessment to guide blood pressure treatment decision-making in moderate- to high-risk individuals, particularly for primary prevention.

Author summary

Why was this study done?

- Hypertension treatment guidelines have traditionally relied primarily on blood pressure levels to guide use of blood pressure-lowering medications.
- Cardiovascular disease (CVD) prevention guidelines, like those for cholesterol management, instead advocate for multivariable CVD risk assessment to guide treatment decisions.
- Simulation studies have modeled the benefits of using CVD risk to guide blood pressure management, but there has not been a direct comparison using clinical trial data with actual outcome events.

What did the researchers do and find?

• We included individual participant data from 11 trials (48,872 participants) of participants in the Blood Pressure Treatment Trialists' Collaboration (BPLTTC).



Cardiology; ACE, angiotensin-converting enzyme; ACEI, angiotensin-converting enzyme inhibitor; ADVANCE, Action in Diabetes and Vascular Disease: Preterax and Diamicron MR Controlled Evaluation; AHA, American Heart Association; BENEDICT, Bergamo Nephrologic Diabetes Complications Trial; BPLTTC, Blood Pressure Lowering Treatment Trialists' Collaboration; CI, confidence interval; CVD, cardiovascular disease; DBP, diastolic blood pressure; HDL, high-density lipoprotein; HOPE-3, Heart Outcomes Prevention Evaluation-3; HOT, Hypertension Optimal Treatment; HYVET, Hypertension in the Very Elderly Trial; PART2, Prevention of Atherosclerosis with Ramipril; PREVENT, Prospective Randomized Evaluation of the Vascular Effects of Norvasc Trial; PROGRESS, Perindopril Protection Against Recurrent Stroke Study; SBP, systolic blood pressure; SCAT, Simvastatin/Enalapril Coronary Atherosclerosis Trial; SPRINT, Systolic Blood Pressure Intervention Trial; SYST-EUR, Systolic Hypertension in Europe.

- We estimated the number of CVD events avoided for a given number of persons treated using CVD risk to determine blood pressure-lowering treatment compared with using systolic blood pressure (SBP) level.
- We compared the two treatment strategies at specific blood pressure thresholds and across the spectrum of CVD risk and SBP levels studied in the BPLTTC.
- We demonstrated that a treatment strategy based on predicted CVD risk could prevent more events and require treatment of fewer persons than one based on SBP level.
- Using predicted CVD risk was particularly beneficial in the primary prevention subgroup.

What do these findings mean?

- These results support using cardiovascular risk assessment to guide blood pressure treatment decisions in moderate- to high-risk individuals.
- The results challenge current paradigms in hypertension management and highlight the merits of using predicted CVD risk to guide blood pressure-lowering treatment intensity.

Introduction

Clinical practice guidelines for hypertension treatment have traditionally relied primarily on blood pressure levels to guide use of blood pressure-lowering medications [1–4]. However, single risk factor levels, like blood pressure, incompletely capture risk. Furthermore, blood pressure-lowering medications provide a fairly consistent relative risk reduction across a range of blood pressure levels, leading to large variations in absolute benefit from blood pressure treatment observed among individuals [5–7].

In contrast to hypertension guidelines, cholesterol treatment guidelines have moved away from single risk factor levels to guide treatment and instead advocate for multivariable absolute cardiovascular disease (CVD) risk assessment to guide treatment decision-making [8–10]. Updates to some cholesterol guidelines have moved even further by eliminating cholesterol goals altogether and identifying CVD risk thresholds to guide clinician-patient risk discussions about statin initiation in primary prevention [8,10].

Recently, an analysis from the Blood Pressure Lowering Treatment Trialists' Collaboration (BPLTTC) demonstrated similar relative risk reductions from blood pressure-lowering medications across a range of predicted risk strata with correspondingly greater absolute risk reductions in those with higher predicted risk [11]. Those results provide support for the role of CVD risk assessment in guiding blood pressure-lowering treatment decisions. Although simulation studies have modeled the benefits of a CVD risk strategy for blood pressure-lowering treatment compared with traditional hypertension guidelines [12,13], to date, there has not been a direct comparison of the two strategies using clinical trial data with actual outcome events. Such evidence is needed to move CVD risk-based treatment strategies into clinical practice. Individual participant data from trials in the BPLTTC provide an ideal opportunity to compare these two treatment strategies across a range of possible treatment thresholds in a group of individuals who were randomly assigned to blood pressure-lowering therapy in a clinical trial setting. In this study, we sought to compare outcomes using a treatment strategy based on predicted CVD risk (CVD risk strategy) with one based on systolic blood pressure (SBP) level (SBP strategy).

Methods

This analysis followed a prespecified protocol that was presented to the BPLTTC Steering committee in April 2013.

Trial eligibility and comparisons

This meta-analysis includes individual participant data from the BPLTTC from 1995 to 2013. Trials were eligible for this analysis if they met the original inclusion criteria for the Collaboration overviews and were part of the subset of trials that randomly allocated participants to blood pressure-lowering drug or placebo or to a more intensive versus less intensive blood pressure drug treatment regimen [14]. Eligible trials were also required to have a minimum of 1,000 patient years of planned follow-up in each randomized group and to have not presented their main results before the Collaboration protocol was finalized in July 1995 [14]. For this analysis, we analysed data from all eligible trials that provided sufficient information to enable absolute CVD risk estimation.

We included the following treatment comparisons: angiotensin-converting enzyme (ACE) inhibitors versus placebo; calcium channel blockers versus placebo; diuretics versus placebo; and more intensive versus less intensive blood pressure-lowering regimens (regardless of drug class). We combined these comparisons to maximize statistical power, since prior analyses have demonstrated that most of the treatment effects were dependent on the amount of blood pressure reduction achieved with few drug-specific effects [15].

Outcomes

We analyzed outcomes prespecified in the original BPLTTC protocol [14]. Our primary outcome was total major CVD events, defined as a composite of stroke (nonfatal stroke or death from cerebrovascular disease), coronary heart disease (nonfatal myocardial infarction or death from coronary heart disease, including sudden death), heart failure (causing death or resulting in admission to hospital), or CVD death.

Cardiovascular risk estimation

We used a multivariable Weibull model previously developed in this dataset from the placebo groups. This model uses age, sex, body mass index, systolic and diastolic blood pressure (DBP), prior blood pressure-lowering treatment, smoking status, diabetes mellitus status, and history of CVD to estimate 5-y CVD risk. Details of the derivation and validation of this model have been previously published [11].

Cardiovascular events avoided by treatment

To estimate the number of CVD events avoided by the CVD risk and SBP strategies, we ranked all eligible participants by decreasing levels of baseline CVD risk and then by decreasing levels of baseline SBP. Next, we considered a potential treatment threshold for each percentile of CVD risk or SBP observed in the dataset. For each strategy, we assumed that all participants with a level above a given threshold would be treated and everyone below would be untreated. We then noted the number of persons above the threshold ("N treated") and the Kaplan-Meier estimated 5-y risks in the control groups above the threshold ("untreated 5-y risk") at each percentile for each treatment strategy. In the subgroup of participants above the threshold, we calculated the relative risk reduction from blood pressure-lowering therapy. For that purpose, we used a one-step meta-analysis approach, fitting Weibull models with shared frailty for each trial, thereby preserving the randomized structure of the trials [16]. To estimate the risk among the treated ("treated 5-y risk"), we applied the obtained hazard ratios to the Kaplan–Meier 5-y risk among control group participants above the threshold using the following formula: treated 5-y risk = $1 - \exp(\text{hazard ratio } \ln(1 - \text{ untreated } 5-\text{ y risk}))$. We thereafter applied this to the total number of persons above the threshold to calculate the number of events avoided with 5 y of treatment: avoided events = *n* treated * (untreated 5-y risk – treated 5-y risk)).

The number of events avoided was graphically presented against the number and proportion of treated persons. We then calculated the areas under the curves to estimate treatment effectiveness (i.e., number of events avoided per persons treated) for each strategy, where greater area reflects more events avoided per persons treated. Areas were calculated as integrals using the trapezoidal rule and were expressed as the ratio of the obtained area to the maximum possible area (maximum number of cardiovascular events avoided multiplied by the maximum number of persons treated); treatment given at random to half the sample without consideration of any treatment thresholds corresponding to an area of 0.5. We used bias-corrected 95% bootstrap estimates from 10,000 repetitions to generate confidence intervals (CIs). We also calculated numbers needed to treat for 5 y to avoid 1 cardiovascular event as 1/(number of events avoided/number of treated persons for 5 y) at each threshold for each treatment strategy.

Secondary analyses

We evaluated the expected outcomes of each treatment strategy in subgroups based on presence or absence of previous blood pressure-lowering treatment, diabetes mellitus, and prevalent CVD. The risk equation was well calibrated in all subgroups (S1 Fig). In a 2-stage meta-analysis combining estimates in complementary pairs of subgroups, heterogeneity of results between subgroups was assessed using I^2 with corresponding 95% CIs. To determine if any differences in cardiovascular events avoided were related to differences in magnitude of SBP reduction, we quantified mean observed SBP reduction with blood pressure-lowering treatment using a mixed-effects generalized linear model with the participant as the unit of analysis and a random intercept for trial. Using these estimates, we standardized hazard ratios for each threshold and for each treatment strategy to a 5 mmHg SBP reduction. Given the strong effect of age on 5-y predicted CVD risk, we performed a sensitivity analysis comparing the CVD risk strategy and SBP strategy with an age-based treatment strategy. For the age-based strategy, we ranked all eligible participants by decreasing levels of age in a manner similar to what was done for baseline SBP in the SBP strategy. Lastly, in a subsample without previous CVD, we performed a sensitivity analysis using the Framingham total CVD risk equation [17]. We recalibrated the published equation to the observed survival rates and risk factor levels (means and proportions) in the BPLTTC dataset, substituting body mass index when trials were missing lipid values.

We used Stata version 14 (Stata Corporation, College Station, TX, United States of America) for all analyses.

Results

Baseline characteristics

We included 11 trials consisting of 47,872 participants (35,671 participants without prevalent CVD) in our analysis (some trials were factorial or included more than two groups) (S5 Fig)

Characteristics	Active/more intensive blood pressure-lowering treatment	Placebo/less intensive blood pressure-lowering treatment	Total
Participants, n	21,021	26,851	47,872
Mean age, y (SD)	65.7 (9.7)	64.7 (9.3)	65.2 (9.5)
Women, <i>n</i> (percentage)	9,614 (46)	12,298 (46)	21,912 (46)
BMI, kg/m ² (SD)	27.6 (4.8)	27.8 (4.8)	27.7 (4.8)
Mean SBP, mmHg (SD)	158 (22)	161 (21)	160 (21)
Mean DBP, mmHg (SD)	91 (13)	94 (13)	93 (13)
Mean total cholesterol, mmol/L (SD)	5.5 (1.5)	5.7 (1.4)	5.6 (1.5)
Mean HDL cholesterol, mmol/L (SD)	1.4 (0.6)	1.4 (0.6)	1.4 (0.6)
Previous antihypertensive treatment, <i>n</i> (percentage)	11,977 (57)	15,073 (56)	27,050 (57)
Current smoking, <i>n</i> (percentage)	3,031 (14)	4,069 (15)	7,100 (15)
Diabetes mellitus, <i>n</i> (percentage)	8,048 (38)	8,225 (31)	16,273 (34)
Previous CVD, <i>n</i> (percentage)	6,051 (29)	6,150 (23)	12,201 (25)

Table 1. Baseline characteristics of participants from the BPLTTC (n = 47,872).

Data obtained from participants in ABCD3_H, ABCD3_N, ADVANCE, BENEDICT1&2, HOT, HYVET, PART2, PREVENT, PROGRESS, SCAT, and SYST-EUR [18-28]. Numbers of patients (*n*) are unbalanced because participants in some trials were not randomly assigned in a 1:1 ratio. To convert from mmol/L to mg/dL for total cholesterol and HDL cholesterol, divide by 0.0259.

Abbreviations: ABCD, Appropriate Blood Pressure Control in Diabetics; ADVANCE, Action in Diabetes and Vascular Disease: Preterax and Diamicron MR Controlled Evaluation; BMI, body mass index; BENEDICT; Bergamo Nephrologic Diabetes Complications Trial; BPLTTC, Blood Pressure Lowering Treatment Trialists' Collaboration; CVD, cardiovascular disease; DBP, diastolic blood pressure; HDL, high-density lipoprotein; HOT, Hypertension Optimal Treatment; HYVET, Hypertension in the Very Elderly Trial; PART2, Prevention of Atherosclerosis with Ramipril; PREVENT, Prospective Randomized Evaluation of the Vascular Effects of Norvasc Trial; PROGRESS, Perindopril Protection Against Recurrent Stroke Study; SCAT, Simvastatin/Enalapril Coronary Atherosclerosis Trial; SBP, systolic blood pressure; SYST-EUR, Systolic Hypertension in Europe.

https://doi.org/10.1371/journal.pmed.1002538.t001

[18–28]. Pooled baseline characteristics of participants are shown in Table 1. Mean systolic and DBP differences between the active/more intense treatment versus placebo/less intense treatment were 5.7/3.2 mmHg (95% CIs 5.5–6.0 mmHg and 3.0–3.3 mmHg, respectively). Baseline characteristics and achieved blood pressure reduction by trial are included in Table 2 and S1 Table.

Cardiovascular events avoided with treatment

There were 3,566 (7.5%) participants who experienced an incident CVD event during a median follow-up of 4.0 y (IQR 1.0, S2 Table). We estimated the number of CVD events avoided over 5 y per person treated according to a CVD risk strategy compared with an SBP strategy (Fig 1). The CVD risk strategy would result in a greater number of events avoided per person treated compared with the SBP strategy. Similarly, for a given number of cardio-vascular events avoided, a smaller proportion of the sample is treated using a CVD risk strategy compared with the SBP strategy. Areas under the curve comparing the two treatment strategies throughout the range of possible thresholds for CVD risk and SBP demonstrated that, on average, a greater number of CVD events would be avoided for a given number of persons treated with the CVD risk strategy compared with the SBP strategy (area under the curve 0.71 [95% CI 0.70–0.72] for the CVD risk strategy versus 0.54 [95% CI 0.53–0.55] for the SBP strategy) (Fig 1).

Characteristic	AB	CD	ADVANCE	BENED	ICT	HOT	HYVET	PART2	PREVENT	PROGRESS	SCAT	SYST-EUR
	Hyper- tensive sample	Normo- tensive sample		ACEI	CCB							
Treatment regimen	More versus less intense	More versus less intense	Perindopril +indapamide versus placebo	Trandolapril versus placebo	Verapamil versus placebo	More versus less intense	Indapamide (+perindopril) versus placebo	Ramipril versus placebo	Amlodipine versus placebo	Perindopril (+indapamide) versus placebo	Enalapril versus placebo	Nifedipine versus placebo
Participants, n	470	479	11,133	453	454	18,776	3,719	617	822	6,074	240	4,635
Mean age, y (SD)	57.8 (8.3)	59.3 (8.3)	65.8 (6.4)	62.0 (8.1)	62.5 (8.2)	61.5 (7.5)	83.5 (3.2)	60.5 (8.1)	57.1 (9.6)	63.9 (9.6)	60.8 (8.7)	69.7 (6.7)
Women, n (%)	153 (33)	218 (46)	4,732 (43)	221 (49)	210 (46)	8,875 (47)	2,255 (61)	114 (18)	163 (20)	1,845 (30)	28 (12)	3,098 (46)
BMI, kg/m2 (SD)	31.7 (5.7)	31.5 (6.0)	28.3 (5.2)	28.9 (4.5)	29.3 (4.6)	28.4 (4.7)	24.7 (3.7)	26.8 (3.6)	28.0 (4.8)	25.7 (3.8)	27.7 (3.9)	27.0 (4.1)
Mean SBP, mmHg (SD)	155 (17)	136 (13)	145 (22)	151 (15)	151 (14)	170 (14)	173 (9)	133 (17)	129 (17)	147 (19)	131 (19)	174 (10)
Mean DBP, mmHg (SD)	98 (7)	84 (3)	81 (11)	88 (8)	88 (8)	105 (3)	91 (9)	79 (10)	(6) 62	86 (11)	79 (10)	85 (6)
Mean total cholesterol level mmol/L (SD)	1	1	5.2 (1.2)	5.4 (1.0)	5.5 (1.0)	6.1 (1.1)	5.3 (1.1)	6.1 (1.1)	5.6 (1.0)	I	5.5 (0.8)	6 (1.2)
Mean HDL cholesterol level mmol/L (SD)			1.3 (0.4)	1.2 (0.3)	1.2 (0.3)	1	1.4 (0.4)	4.1 (0.9)	1.2 (0.3)	1	1.0 (0.2)	1.4 (0.5)
Previous antihypertensive treatment, n (%)	209 (44)	149 (31)	7,651 (69)	453 (100)	454 (100)	9,868 (53)	2,017 (54)	337 (55)	550 (67)	3,051 (50)	136 (57)	2,175 (47)
Smokers, n (%)	65 (14)	63 (13)	1,681 (15)	63 (14)	52 (11)	2,979 (16)	246 (7)	100 (16)	203 (25)	1,274 (21)	36 (15)	338 (7)
Diabetes mellitus, n (%)	470 (100)	479 (100)	11,133 (100)	453 (100)	454(100)	1,501 (8)	369 (10)	51 (8)	98 (12)	759 (12)	23 (10)	483 (10)
Previous cardiovascular disease, n (%)	50 (11)	39 (8)	3,588 (32)	0 (0)	(0) 0	313 (2)	358 (10)	457 (74)	822 (100)	6,074 (100)	218 (91)	282 (6)
10-y total CVD risk, % (SD)		1	21 (12)	22 (11)	23 (11)	1	26 (13)	4 (3)	10 (7)	1	12 (8)	19 (11)
10-y CHD risk, % (SD)	1	1	15 (9)	15 (8)	15 (8)	1	17 (10)	3 (2)	7 (5)	I	9 (5)	12 (8)
10-y total CVD and (Abbreviations: ABC Diamicron MR Cont diastolic blood presst	CHD risks (D, Appropi trolled Evalu ure; HDL, h	letermined u riate Blood P uation; BENI igh-density l	asing the Framing ressure Control i EDICT, Bergamo lipoprotein; HOT	gham Heart Stud n Diabetics; ACF Nephrologic Dia 7, Hypertension C	y equations. T II, angiotensir abetes Compli Dptimal Treati	Fo convert f n-convertin ications Trii ment; HYV	rom mmol/L to m g enzyme inhibito al; BMI, body mas: TT, Hypertension	g/dL for total r; ADVANCF s index; CHD in the Very F	cholesterol and 3, Action in Dia 1, coronary hear 3lderly Trial; PA	(HDL cholesterol, c betes and Vascular t disease; CVD, car r12, Prevention of	livide by 0.02 Disease: Prete diovascular d	59. rrax and sease; DBP, sis with
Ramipril; PREVENT	., Prospectiv	<i>i</i> e Randomiz	red Evaluation of	the Vascular Effe	ets of Norvas	c Trial; PR(JGRESS , Perindop	ril Protection	n Against Recur	rent Stroke Study; S	BP, systolic b	lood pressure;

https://doi.org/10.1371/journal.pmed.1002538.t002

SCAT, Simvastatin/Enalapril Coronary Atherosclerosis Trial; SYST-EUR, Systolic Hypertension in Europe.

Table 2. Characteristics of included trials.



Fig 1. Effects of CVD risk and SBP treatment strategies on absolute number of cardiovascular events prevented over 5 y. Expected cardiovascular events avoided over 5 y of treatment as a function of number of persons and proportion of sample treated using a CVD risk strategy (in green) and an SBP strategy (in red). Numbers associated with each curve represent the specific CVD risk level (percentage 5-y CVD risk) or SBP (mmHg) at the treatment threshold. Areas under the curve are expressed as the ratio of the obtained area to the maximum possible area (maximum number of cardiovascular events avoided multiplied by the maximum number of participants treated) with bias-corrected 95% bootstrap CIs from 10,000 repetitions in parentheses. Larger areas represent more events avoided per persons treated. CI, confidence interval, CVD, cardiovascular disease; SBP, systolic blood pressure.

https://doi.org/10.1371/journal.pmed.1002538.g001

The number needed to treat for 5 y to prevent 1 CVD event was lower with the CVD risk strategy across a broad range of thresholds until overlapping with the SBP strategy at the 80th percentile treatment rate (Fig 2). Hence, the CVD risk strategy was superior to the SBP strategy in terms of identifying the persons with the highest absolute treatment benefit across a broad range of plausible treatment thresholds.

More specifically, we selected three commonly proposed SBP treatment thresholds for direct comparison with the CVD risk strategy (Tables <u>3</u> and <u>4</u>). Compared with treatment at an SBP threshold of 150 mmHg, a CVD risk threshold would require the treatment of 29% (95% CI 26%–31%) fewer individuals to prevent the same number of CVD events (Table <u>3</u>). Alternatively, it could prevent 16% (95% CI 14%–18%) more CVD events for the same number of persons treated (Table <u>4</u>). Similarly, compared with an SBP threshold of 160 mmHg, using a CVD risk threshold would require treatment of 35% (95% CI 50%–24%) fewer persons to prevent the same number of CVD events, or it could prevent 38% (95% CI 29%–40%) more CVD events for the same number of treated persons. Results at an SBP threshold of 140 mmHg were imprecise due to the nature of our comparisons. At this threshold, the CVD risk strategy would prevent 3% (95% CI 1.5%–5%) more CVD events for the same number of events prevented, although this result was not statistically significant.



Fig 2. Numbers needed to treat for 5 y to avoid 1 cardiovascular event according to CVD risk and SBP treatment strategies. Numbers needed to treat for 5 y to avoid 1 cardiovascular event as a function of number of persons and proportion of sample treated using a CVD risk strategy (in green) and an SBP strategy (in red). Numbers associated with each curve represent the specific CVD risk level (percentage 5-y CVD risk) or SBP (mmHg) at the treatment threshold. CVD, cardiovascular disease; SBP, systolic blood pressure.

https://doi.org/10.1371/journal.pmed.1002538.g002

Table 3. Persons needed to treat to avoid the same number of cardiovascular events over 5 y using selected S	SBP
thresholds and corresponding CVD risk thresholds. The BPLTTC ($n = 47,872$).	

			SBP threshold	
		140 mmHg	150 mmHg	160 mmHg
Persons and proportions treated using an SBP threshold	п	39,231 (39,231 to 39,344)	33,891 (33,551 to 34,460)	27,039 (26,245 to 27,882)
	percentage	81.9 (81.9 to 82.2)	70.8 (70.1 to 72.0)	56.5 (54.8 to 58.2)
Persons and proportions treated using a CVD risk threshold selected to avoid the same number of	п	37,730 (34,343 to 42,073)	24,225 (23,743 to 25,660)	17,484 (14,108 to 21,333)
cardiovascular events	percentage	78.8 (71.7 to 87.9)	50.6 (49.6 to 53.6)	36.5 (29.5 to 44.6)
Difference in persons and relative difference in proportions treated using SBP and CVD risk thresholds selected to avoid the same number of	n	-1,501 (-4,888 to +2,842)	-9,667 (-10,717 to -8,504)	-9,556 (-13,095 to -6,603)
cardiovascular events	percentage	-3.8 (-12.5 to +7.2)	-28.5 (-31.1 to -25.6)	-35.3 (-49.9 to -24.2)

Estimates with 10,000 repetitions bias-corrected bootstrap with 95% CIs in parentheses. Negative sign represents smaller proportion or lesser number. Positive sign represents bigger proportion or greater number.

Abbreviations: BPLTTC, Blood Pressure Lowering Treatment Trialists' Collaboration; CI, confidence interval; CVD, cardiovascular disease; SBP, systolic blood pressure.

https://doi.org/10.1371/journal.pmed.1002538.t003

			SBP threshold	
		140 mmHg	150 mmHg	160 mmHg
		<i>n</i> = 39,231	<i>n</i> = 33,891	<i>n</i> = 27,039
CVD events avoided using an SBP threshold	n	821 (819 to 821)	698 (692 to 716)	557 (544 to 599)
CVD events avoided using a CVD risk threshold selected to achieve the same number of persons treated	n	847 (833 to 860)	809 (795 to 822)	767 (755 to 776)
Difference and relative difference in CVD events avoided using SBP and CVD risk thresholds selected	n	+26 (+12 to +41)	+111 (+97 to +127)	+210 (175 to 222)
to achieve the same number of persons treated	percentage	+3.1 (+1.5 to +5.0)	+15.8 (+13.7 to +18.3)	+37.6 (+28.8 to +40.2)

Table 4. Cardiovascular events avoided over 5 y for the same number of persons treated using selected SBP thresholds and corresponding CVD risk thresholds. The BPLTTC (n = 47,872).

Estimates with 10,000 repetitions bias-corrected bootstrap with 95% CIs in parentheses. Negative sign represents smaller proportion or lesser number. Positive sign represents greater number.

Abbreviations: BPLTTC, Blood Pressure Lowering Treatment Trialists' Collaboration; CI, confidence interval; CVD, cardiovascular disease; SBP, systolic blood pressure.

https://doi.org/10.1371/journal.pmed.1002538.t004

Secondary analyses

Results were similar in subgroups with and without prior blood pressure-lowering medication use, without diabetes mellitus, and without prevalent CVD (Fig 3). For those with baseline diabetes mellitus, the CVD risk strategy did not appear to be superior, while for those with prior CVD, the SBP strategy appeared to be best (Fig 3).

Mean achieved SBP reductions and observed relative risk reductions for both the SBP strategy and CVD risk strategy were similar across the range of possible CVD risk and SBP thresholds (Fig 4 and Fig 5).

Results were similar for analyses standardized to a 5-mmHg SBP reduction (S2 Fig and S3 Fig), but differences were smaller. Furthermore, analyses comparing the two treatment strategies with an age-based treatment strategy confirmed that the CVD risk strategy remained superior in terms of events avoided per person treated compared with the SBP strategy (difference in areas under the curves 0.17 [95% CI 0.15–0.19]) and age-based strategy (difference in areas under the curves 0.13 [95% CI 0.09–0.14]) (S4 Fig).

For analyses using the Framingham total CVD risk equation, a greater number of CVD events would be avoided for a given number of persons treated with the CVD risk strategy compared with the SBP strategy (area under the curve 0.66 [95% CI 0.65–0.72] for CVD risk strategy versus 0.59 [95% CI 0.57–0.61] for the SBP strategy), although differences were smaller than in analyses using the internally derived risk prediction equation.

Discussion

This analysis of nearly 50,000 persons studied in clinical trials demonstrated that a blood pressure-lowering treatment strategy based on predicted CVD risk could prevent more events for the same number of persons treated compared with a strategy based on SBP levels. The benefit of the CVD risk strategy was particularly evident at higher SBP thresholds and for persons without prevalent CVD or diabetes mellitus.

The superiority of the CVD risk strategy relative to the SBP strategy can be explained by recent lessons from blood pressure epidemiology. First, the relative treatment benefit from blood pressure-lowering medications is fairly consistent across different blood pressure levels,

Subgroup

Avoided events difference (95% CI)



Fig 3. Performance of CVD risk and SBP treatment strategies in subgroups defined at baseline. Estimates represent differences in areas under the curve between CVD risk and SBP treatment strategies for the number of cardiovascular events avoided per persons treated in each subgroup, defined at baseline. Bias-corrected 95% bootstrap CIs from 10,000 repetitions in parentheses. Heterogeneity for the dataset determined for meta-analyses of two complementary strata at a time was assessed using I² and corresponding 95% CIs. BP, blood pressure; CI, confidence interval, CVD, cardiovascular disease; SBP, systolic blood pressure.

https://doi.org/10.1371/journal.pmed.1002538.g003

including nonhypertensive levels [6,7]. Second, the absolute risk of an individual at a given blood pressure level can vary as much as 20-fold based on the presence of other vascular risk factors such as age, sex, dyslipidemia, and diabetes mellitus [5,29]. Third, a previous metaanalysis of individual participant data from the BPLTTC has shown that the relative benefit from blood pressure-lowering treatment is similar across risk strata and that, therefore, the absolute benefit from blood pressure-lowering treatment is greater in those with higher risk [11]. Therefore, the expected absolute risk reduction achieved with blood pressure-lowering treatment is better determined by the combination of risk factors contributing to CVD risk rather than an isolated blood pressure level. In the present study, we quantified the benefits of such a risk-based strategy and determined which groups of patients might experience those benefits.

Indirect comparisons of CVD risk and SBP treatment strategies have previously been performed using observational cohorts and modeled treatment effects. These analyses suggest that treatment strategies based on absolute risk could prevent more cardiovascular events, save more quality-adjusted life-years, use fewer medications, and lower overall costs compared with treatment based on blood pressure level [12,13,30,31]. Our results confirm, quantify, and extend these findings in a large group of persons who actually received blood pressure-lowering medications in randomized clinical trials across a broad range of CVD risk and SBP levels.

It should be noted that this is a proof-of-principle study. The results of the two strategies will, by design, appear to converge at the upper and lower ends of the CVD risk and SBP distributions that are determined by the composition of this sample and findings at the ends of the distributions are uncertain. Therefore, although the results of the present study illustrate the



Fig 4. Achieved SBP reduction according to CVD risk and SBP treatment strategies. Achieved SBP reduction as a function of number of persons and proportion of sample treated using a CVD risk strategy (in green) and an SBP strategy (in red). Numbers associated with each curve represent the specific CVD risk level (percentage 5-y CVD risk) or SBP (mmHg) at the treatment threshold. CVD, cardiovascular disease; SBP, systolic blood pressure.

https://doi.org/10.1371/journal.pmed.1002538.g004

principle that a CVD risk strategy outperforms an SBP strategy over a broad range of SBPs and CVD risks, there is a need for further studies elucidating the differences between these two strategies at the lower ends of the blood pressure and CVD risk distributions, particularly because these are values that are taken into consideration when discussing thresholds for therapeutic intervention. The Systolic Blood Pressure Intervention Trial (SPRINT) and the Heart Outcomes Prevention Evaluation-3 (HOPE-3) study provide data that complement our analyses and support these results [32,33]. In SPRINT, blood pressure reduction in a high-risk group (annual event rate of 2.2%/y) below conventional targets led to a 25% reduction in cardiovascular events and a 27% reduction in all-cause deaths [32]. In contrast, HOPE-3 did not demonstrate a cardiovascular benefit from blood pressure reduction in a lower-risk group (annual event rate 0.8%/y) where the majority of participants had SBP below conventional targets [33].

In subgroup analyses, we found no evidence of a benefit for a CVD risk strategy compared with SBP strategy in patients with diabetes mellitus or existing CVD. This may reflect that "high risk" is already captured in these groups, independent of the predicted CVD risk estimate. In addition, the relatively weaker performance of CVD risk prediction among persons with existing CVD could also be explained by phenomena like index event bias [34] or observations that consequences of a cardiovascular event are often stronger predictors of a subsequent event than traditional risk factors [35–37]. Given the more uniformly high-risk status of persons with prevalent CVD or diabetes mellitus, a "treat-all" strategy may be better than selective treatment based on predicted CVD risk or blood pressure level. Thus, utilizing predicted CVD risk to guide treatment decisions may be best suited in primary prevention for individuals without diabetes mellitus or CVD where baseline risks and expected benefits from blood



Fig 5. Relative risk reductions by blood pressure-lowering treatment, according to CVD risk and SBP treatment strategies. Hazard ratios of major cardiovascular events by blood pressure-lowering treatment, as a function of number of persons and proportion of sample treated using a CVD risk strategy (in green) and an SBP strategy (in red). Numbers associated with each curve represent the specific CVD risk level (percentage 5-y CVD risk) or SBP (mmHg) at the treatment threshold. Hazard ratios determined from persons above each treatment threshold. CVD, cardiovascular disease; SBP, systolic blood pressure.

https://doi.org/10.1371/journal.pmed.1002538.g005

pressure-lowering treatment are more heterogeneous. This study, therefore, provides clinical guidance for treatment decisions in a broad segment of the general population.

Clinical implications

Our results support the use of absolute risk assessment in guiding blood pressure-lowering treatment decisions. Although risk-based treatment has been a cornerstone of cholesterol management [9,10], blood pressure treatment guidelines like those from the US have historically emphasized blood pressure thresholds and targets [1,38]. The present study challenges this paradigm and instead highlights the merits of using predicted CVD risk to guide intensity of blood pressure-lowering with medications, a framework recently embraced by the 2017 American College of Cardiology (ACC)/American Heart Association (AHA) Hypertension guidelines [39]. The limited trial experience with treatment to SBP below 120 mmHg necessarily limits the scope of an entirely risk-based treatment strategy. Nevertheless, our results support the principle that treatment decisions that are based on absolute CVD risk compared with blood pressure-lowering treatment, especially in primary prevention. This assessment can form the basis for a shared clinician-patient discussion to contextualize expected benefits and harms of blood pressure treatment with individual values and preferences to personalize treatment decisions.

PLOS | MEDICINE

Strengths and limitations

A key strength of this analysis is the use of a high-quality dataset of persons who were actually treated with blood pressure-lowering medications as part of a randomized clinical trial with rigorous follow-up and adjudicated outcomes (S5 Fig). Furthermore, our calculation of relative risk reductions from blood pressure-lowering treatment at each potential threshold preserves the randomized structure of the trials and maximizes the information within this dataset. There are, however, important limitations to acknowledge in this analysis. First, most trial participants were hypertensive at baseline, were receiving background blood pressure-lowering therapy, and had an estimated 5-y CVD risk greater than 5%. Therefore, the majority of the difference between the two strategies was seen among participants with SBP 150 to 170 mmHg or 5-y CVD risk 7.5% to 15%. As such, we were unable to detect a difference between the two treatment strategies at a 5-y CVD risk level of less than 5% or at an SBP threshold of 140 mmHg. The inability of this method to provide information at the ends of the risk and blood pressure distributions motivates future studies in other samples that include individuals with lower blood pressures or risk. Second, trials were of relatively short duration, and therefore, these results do not account for the potential long-term benefits (or harms) of sustained blood pressure management over the course of a lifetime [40]. Third, the utility of a CVD risk strategy depends on the performance of the prediction algorithm employed. For this study, we chose a previously validated, internally derived prediction algorithm to optimize model performance. Thus, these results have high internal but unknown external validity. To account for this limitation, we performed sensitivity analyses using the Framingham total CVD risk equation. The results, while qualitatively similar, had a lower area under the curve, suggesting worse performance of the CVD risk strategy using an externally derived prediction algorithm. It should be noted that modifying the used risk equation to prediction of 10- instead of 5-y risks would yield the same ranking of participants and hence produce the same relative results. Finally, this is a post-hoc analysis of data from clinical trials that had different objectives and entry criteria and with numerical results driven by the composition of the study sample. Therefore, prospective validation in a new trial would be the ideal way to confirm our findings, but given the large number of participants required, such a trial is unlikely to be performed. These limits notwithstanding, these results should serve as a proof of principle of the relative merits of a CVD risk blood pressure treatment strategy in a high-quality dataset of persons treated with blood pressure-lowering medications.

Conclusion

In conclusion, this individual participant data analysis of blood pressure-lowering trial participants supports the principle that a treatment strategy based on predicted CVD risk, compared with one based on SBP levels, would result in prevention of more cardiovascular events for the same number of treated persons across a wide range of potential treatment thresholds. These results support use of cardiovascular risk assessment to guide blood pressure-lowering treatment decision-making in moderate- to high-risk individuals, especially in primary prevention settings.

Supporting information

S1 Table. Mean blood pressure reductions in included trials. (DOCX)

S2 Table. Number of events by trial included in analysis. (DOCX)

S1 Fig. Observed versus expected 5-y risks of cardiovascular events in prespecified subgroups. Calibration of the internally derived CVD risk prediction equation in the prespecified subgroups. CVD, cardiovascular disease. (PDF)

S2 Fig. Effects of CVD risk and SBP treatment strategies on absolute number of cardiovascular events prevented, standardized to a 5-mmHg SBP reduction. Expected cardiovascular events avoided over 5 y as a function of number of persons and proportion of sample treated using a CVD risk strategy (in green) and an SBP strategy (in red), standardized to a 5-mmHg SBP reduction. Numbers associated with each curve represent the specific CVD risk level (percentage 5-y CVD risk) or SBP (mmHg) at the treatment threshold. Areas under the curve are expressed as the ratio of the obtained area to the maximum possible area (maximum number of cardiovascular events avoided multiplied by the maximum number of persons treated) with bias-corrected 95% bootstrap CIs from 10,000 repetitions in parentheses. Larger areas represent more avoidable events avoided per persons treated. CI, confidence interval; CVD, cardiovascular disease; SBP, systolic blood pressure. (PDF)

S3 Fig. Numbers needed to treat for 5 y to avoid 1 cardiovascular event according to CVD risk and SBP treatment strategies, standardized to a 5-mmHg SBP reduction. Numbers needed to treat for 5 y to avoid 1 cardiovascular event as a function of number of persons and proportion of sample treated using a CVD risk strategy (in green) and an SBP strategy (in red), standardized to a 5-mmHg SBP reduction. Numbers associated with each curve represent the specific CVD risk level (percentage 5-y CVD risk) or SBP (mmHg) at the treatment threshold. CVD, cardiovascular disease; SBP, systolic blood pressure. (PDF)

S4 Fig. Effects of CVD risk-, SBP-, and age-based treatment strategies on absolute number of cardiovascular events prevented. Expected cardiovascular events avoided over 5 y as a function of number of persons and proportion of sample treated using a CVD risk strategy (in green), an SBP strategy (in red), and an age-based strategy (in blue). Numbers associated with each curve represent the specific CVD risk (percentage 5-y CVD risk), SBP (mmHg), or age (y) level at that treatment threshold. CVD, cardiovascular disease; SBP, systolic blood pressure. (PDF)

S5 Fig. Flow chart. (PDF)

Acknowledgments

The Blood Pressure Lowering Treatment Trialists' Collaboration

Steering committee

Kazem Rahimi (chair), Koon Teo, Barry R. Davis, John Chalmers, Carl Pepine

Collaborating Trialists

L. Agodoa (AASK [African-American Study of Kidney Disease and Hypertension]), F. W. Asselbergs (PREVEND-IT [Prevention of Renal and Vascular End- stage Disease Intervention Trial], Renal analyses), N. Beckett (HYVET [Hypertension in the Very Elderly Trial]), E. Berge (VALUE trial [Valsartan Antihypertensive Long-term Use Evaluation trial]), H. Black (CONVINCE [Controlled Onset Verapamil Investigation of Cardiovascular End Points]), B. Brenner (RENAAL [Reduction of Endpoints in NIDDM with the Angiotensin II Antagonist Losartan]), F. P. J. Brouwers (PREVEND-IT), M. Brown (INSIGHT [International Nifedipine GITS Study: Intervention as a Goal in Hypertension]), C. J. Bulpitt (HYVET), B. Byington (PREVENT [Prospective Randomized Evaluation of the Vascular Effects of Norvasc Trial]), J. Chalmers (ADVANCE [Action in Diabetes and Vascular Disease: Preterax and Diamicron MR Controlled Evaluation]), J. Cutler (ALLHAT [Antihypertensive and Lipid-Lowering Treatment to Prevent Heart Attack Trial]), B. Davis (ALLHAT), R. Estacio (ABCD [Appropriate Blood Pressure Control in Diabetes]), R. Fagard (SYST-EUR [SYSTtolic Hypertension in EURope]), K. Fox (EUROPA [European trial on Reduction Of cardiac events with Perindopril among patients with stable coronary Artery disease]), T. Fukui (CASE-J [Candesartan Antihypertensive Survival Evaluation in Japan]), Y. Imai (HOMED-BP [Hypertension Objective Treatment Based on Measurement by Electrical Devices of Blood Pressure]), R. Holman (UKPDS [UK Prospective Diabetes Study]), M. Ishii (JATOS [Japanese Trial to Assess Optimal Systolic Blood Pressure in Elderly Hypertensive Patients], JMIC-B [Japan Multicenter Investigation for Cardiovascular Diseases-B]), S. Julius (VALUE), Y. Kanno (ECOST [The Effectiveness and Cost of ICD follow-up Schedule with Telecardiology]), S. Kjeldsen (VALUE, LIFE [Losartan Intervention For Endpoint reduction in hypertension]), K. Kuramoto (NICS-EH [National Intervention Cooperative Study in Elderly Hypertensives]), J. Lanke (STOP2 [Swedish Trial in Old Patients with Hypertension-2], NORDIL [Nordic Diltiazem]), E. Lewis (IDNT [Irbesartan Diabetic Nephropathy Trial]), M. Lievre (DIABHYCAR [Non-insulin-dependent diabetes, hypertension, microalbuminuria or proteinuria, cardiovascular events, and ramipril study]), L. H. Lindholm (CAPPP [Captopril Prevention Project], STOP2, NORDIL), L. Lisheng (FEVER [Felodipine Event Reduction]), J. Lubsen (ACTION [A Coronary disease Trial Investigating Outcome with Nifedipine GITS]), S. Lueders (MOSES [The Morbidity and Mortality After Stroke, Eprosartan Compared With Nitrendipine for Secondary Prevention]), E. Malacco (SHELL [Systolic Hypertension in the Elderly: Lacidipine Long-Term study]), G. Mancia (SHELL), S. MacMahon (ADVANCE), M. Matsuzaki (COPE [The Combination Therapy of Hypertension to Prevent Cardiovascular Events]), M. Mehlum (VALUE), S. Nissen (CAMELOT [Comparison of Amlodipine vs Enalapril to Limit Occurrences of Thrombosis]), H. Ogawa (HIJ-CREATE), T. Ogihara (CASE-J), T. Ohkubo (HOMED-BP), C. Palmer (INSIGHT), A. Patel (ADVANCE), C. Pepine (INVEST [International Verapamil SR-Trandolapril Study]), M Pfeffer (PEACE [Prevention of Events With Angiotensin- Converting Enzyme Inhibition]), B. Pitt (QUIET [QUinapril Ischemic Event Trial]), N. R. Poulter (ASCOT [Anglo-Scandinavian Cardiac Outcomes Trial]), H. Rakugi (VALISH/CASE-J), G. Remuzzi (BENEDICT [BErgamo NEphrologic Dlabetes Complications Trial]), P. Ruggenenti (BENEDICT), T. Saruta (CASE-J), J. Schrader (MOSES), R. Schrier (ABCD), P. Sever (ASCOT), P. Sleight (CONVINCE, HOPE [Heart Outcomes Prevention Evaluation], TRAN-SCEND [Telmisartan Randomised AssessmeNt Study in ACE iNtolerant subjects with cardiovascular Disease], ONTARGET [Ongoing Telmisartan Alone and in Combination with Ramipril Global Endpoint Trial]), J. Staessen (SYST-EUR [Systolic Hypertension in Europe]), H. Suzuki (ECOST [Efficacy of Candesartan on Outcome in Saitama Trial]), K. Teo (SCAT [Simvastatin/Enalapril Coronary Atherosclerosis Trial]), L. Thijs (Syst-Eur), K. Ueshima (VALISH/CASE-J), S. Umemoto (COPE), W. H. van Gilst (PREVEND-IT/Renal analyses), P. Verdeccia (Cardio-Sis), G. Viberti (ROADMAP [Randomised Olmesartan and Diabetes Microalubuminuria Prevention]), J. Wang (NICOLE [NIsoldipine in COronary artery disease in LEuven]), L. Wing (ANBP2 [The Second Australian National Blood Pressure Study]), Y. Yui (JMIC-B), S. Yusuf (HOPE, ONTARGET, TRANSCEND), A. Zanchetti (SCOPE [Study on COgnition and Prognosis in the Elderly], VHAS [Verapamil in Hypertension and

Atherosclerosis Study], ELSA [European Lacidipine Study on Atherosclerosis], HOT [Hypertension Optimal Treatment]).

Other members

C. Baigent, R. Collins, D. de Zeeuw, L. Hansson (deceased), J. Kostis, B. Neal, A. Perez, V. Perkovic, P. Poole-Wilson (deceased), M. Rahman, W. J. Remme, A. Rodgers, J. Sundström, F. Turnbull.

The views expressed in this article are those of the authors and do not necessarily represent the views of the National Heart, Lung, and Blood Institute; National Institutes of Health; or the United States Department of Health and Human Services.

Author Contributions

Conceptualization: Kunal N. Karmali, Johan Sundström.

Data curation: Koon Teo, Barry R. Davis, John Chalmers, Carl Pepine, Johan Sundström.

- Investigation: Kunal N. Karmali, Donald M. Lloyd-Jones, Joep van der Leeuw, David C. Goff, Jr., Salim Yusuf, Alberto Zanchetti, Paul Glasziou, Rodney Jackson, Mark Woodward, Anthony Rodgers, Bruce C. Neal, Eivind Berge, Johan Sundström.
- Methodology: Kunal N. Karmali, Donald M. Lloyd-Jones, Joep van der Leeuw, David C. Goff, Jr., Salim Yusuf, Alberto Zanchetti, Paul Glasziou, Rodney Jackson, Mark Woodward, Anthony Rodgers, Bruce C. Neal, Eivind Berge, Koon Teo, Barry R. Davis, John Chalmers, Carl Pepine, Johan Sundström.

Project administration: Kazem Rahimi.

Visualization: Johan Sundström.

Writing – review & editing: Kunal N. Karmali, Donald M. Lloyd-Jones, Joep van der Leeuw, David C. Goff, Jr., Salim Yusuf, Alberto Zanchetti, Paul Glasziou, Rodney Jackson, Mark Woodward, Anthony Rodgers, Bruce C. Neal, Eivind Berge, Koon Teo, Barry R. Davis, John Chalmers, Carl Pepine, Kazem Rahimi, Johan Sundström.

References

- James PA, Oparil S, Carter BL, Cushman WC, Dennison-Himmelfarb C, Handler J, et al. 2014 evidence-based guideline for the management of high blood pressure in adults: report from the panel members appointed to the Eighth Joint National Committee (JNC 8). JAMA. 2014; 311(5):507–20. Epub 2013/12/20. https://doi.org/10.1001/jama.2013.284427 PMID: 24352797.
- Krause T, Lovibond K, Caulfield M, McCormack T, Williams B. Management of hypertension: summary of NICE guidance. BMJ. 2011; 343:d4891. Epub 2011/08/27. https://doi.org/10.1136/bmj.d4891 PMID: 21868454.
- Weber MA, Schiffrin EL, White WB, Mann S, Lindholm LH, Kenerson JG, et al. Clinical practice guidelines for the management of hypertension in the community: a statement by the American Society of Hypertension and the International Society of Hypertension. J Clin Hypertens (Greenwich). 2014; 16 (1):14–26. Epub 2013/12/18. https://doi.org/10.1111/jch.12237 PMID: 24341872.
- Mancia G, Fagard R, Narkiewicz K, Redon J, Zanchetti A, Bohm M, et al. 2013 ESH/ESC guidelines for the management of arterial hypertension: the Task Force for the Management of Arterial Hypertension of the European Society of Hypertension (ESH) and of the European Society of Cardiology (ESC). Eur Heart J. 2013; 34(28):2159–219. Epub 2013/06/19. https://doi.org/10.1093/eurheartj/eht151 PMID: 23771844.
- Jackson R, Lawes CM, Bennett DA, Milne RJ, Rodgers A. Treatment with drugs to lower blood pressure and blood cholesterol based on an individual's absolute cardiovascular risk. Lancet. 2005; 365 (9457):434–41. Epub 2005/02/01. https://doi.org/10.1016/S0140-6736(05)17833-7 PMID: 15680460.

- Ettehad D, Emdin CA, Kiran A, Anderson SG, Callender T, Emberson J, et al. Blood pressure lowering for prevention of cardiovascular disease and death: a systematic review and meta-analysis. Lancet. 2016; 387(10022):957–67. Epub 2016/01/03. https://doi.org/10.1016/S0140-6736(15)01225-8 PMID: 26724178.
- Thomopoulos C, Parati G, Zanchetti A. Effects of blood pressure lowering on outcome incidence in hypertension: 3. Effects in patients at different levels of cardiovascular risk—overview and meta-analyses of randomized trials. J Hypertens. 2014; 32(12):2305–14. Epub 2014/09/27. <u>https://doi.org/10.1097/HJH.000000000000880 PMID: 25259548</u>.
- Joint British Societies' consensus recommendations for the prevention of cardiovascular disease (JBS3). Heart. 2014; 100 Suppl 2:ii1–ii67. Epub 2014/03/29. https://doi.org/10.1136/heartjnl-2014-305693 PMID: 24667225.
- Catapano AL, Graham I, De Backer G, Wiklund O, Chapman MJ, Drexel H, et al. 2016 ESC/EAS Guidelines for the Management of Dyslipidaemias. Eur Heart J. 2016; 37(39):2999–3058. Epub 2016/08/28. https://doi.org/10.1093/eurheartj/ehw272 PMID: 27567407.
- Stone NJ, Robinson JG, Lichtenstein AH, Bairey Merz CN, Blum CB, Eckel RH, et al. 2013 ACC/AHA guideline on the treatment of blood cholesterol to reduce atherosclerotic cardiovascular risk in adults: a report of the American College of Cardiology/American Heart Association Task Force on Practice Guidelines. Circulation. 2014; 129(25 Suppl 2):S1–45. Epub 2013/11/14. https://doi.org/10.1161/01.cir. 0000437738.63853.7a PMID: 24222016.
- Sundstrom J, Arima H, Woodward M, Jackson R, Karmali K, Lloyd-Jones D, et al. Blood pressure-lowering treatment based on cardiovascular risk: a meta-analysis of individual patient data. Lancet. 2014; 384(9943):591–8. Epub 2014/08/19. https://doi.org/10.1016/S0140-6736(14)61212-5 PMID: 25131978.
- Eddy DM, Adler J, Patterson B, Lucas D, Smith KA, Morris M. Individualized guidelines: the potential for increasing quality and reducing costs. Ann Intern Med. 2011; 154(9):627–34. Epub 2011/05/04. <u>https:// doi.org/10.7326/0003-4819-154-9-201105030-00008 PMID: 21536939.</u>
- Sussman J, Vijan S, Hayward R. Using benefit-based tailored treatment to improve the use of antihypertensive medications. Circulation. 2013; 128(21):2309–17. Epub 2013/11/06. https://doi.org/10.1161/ CIRCULATIONAHA.113.002290 PMID: 24190955; PubMed Central PMCID: PMCPmc4026201.
- Protocol for prospective collaborative overviews of major randomized trials of blood-pressure-lowering treatments. World Health Organization-International Society of Hypertension Blood Pressure Lowering Treatment Trialists' Collaboration. J Hypertens. 1998; 16(2):127–37. Epub 1998/04/16. PMID: 9535138.
- Neal B, MacMahon S, Chapman N. Effects of ACE inhibitors, calcium antagonists, and other bloodpressure-lowering drugs: results of prospectively designed overviews of randomised trials. Blood Pressure Lowering Treatment Trialists' Collaboration. Lancet. 2000; 356(9246):1955–64. Epub 2000/12/29. PMID: 11130523.
- Reilly BM, Evans AT. Translating clinical research into clinical practice: impact of using prediction rules to make decisions. Ann Intern Med. 2006; 144(3):201–9. Epub 2006/02/08. PMID: 16461965.
- D'Agostino RB Sr., Vasan RS, Pencina MJ, Wolf PA, Cobain M, Massaro JM, et al. General cardiovascular risk profile for use in primary care: the Framingham Heart Study. Circulation. 2008; 117(6):743– 53. Epub 2008/01/24. https://doi.org/10.1161/CIRCULATIONAHA.107.699579 PMID: 18212285.
- Estacio RO, Jeffers BW, Hiatt WR, Biggerstaff SL, Gifford N, Schrier RW. The effect of nisoldipine as compared with enalapril on cardiovascular outcomes in patients with non-insulin-dependent diabetes and hypertension. N Engl J Med. 1998; 338(10):645–52. Epub 1998/03/05. <u>https://doi.org/10.1056/</u> NEJM199803053381003 PMID: 9486993.
- Schrier RW, Estacio RO, Esler A, Mehler P. Effects of aggressive blood pressure control in normotensive type 2 diabetic patients on albuminuria, retinopathy and strokes. Kidney Int. 2002; 61(3):1086–97. Epub 2002/02/19. https://doi.org/10.1046/j.1523-1755.2002.00213.x PMID: 11849464.
- Ruggenenti P, Fassi A, Ilieva AP, Bruno S, Iliev IP, Brusegan V, et al. Preventing microalbuminuria in type 2 diabetes. N Engl J Med. 2004; 351(19):1941–51. Epub 2004/11/02. <u>https://doi.org/10.1056/</u> NEJMoa042167 PMID: 15516697.
- Hansson L, Zanchetti A, Carruthers SG, Dahlof B, Elmfeldt D, Julius S, et al. Effects of intensive bloodpressure lowering and low-dose aspirin in patients with hypertension: principal results of the Hypertension Optimal Treatment (HOT) randomised trial. HOT Study Group. Lancet. 1998; 351(9118):1755–62. PMID: 9635947.
- Beckett NS, Peters R, Fletcher AE, Staessen JA, Liu L, Dumitrascu D, et al. Treatment of hypertension in patients 80 years of age or older. N Engl J Med. 2008; 358(18):1887–98. Epub 2008/04/02. https:// doi.org/10.1056/NEJMoa0801369 PMID: 18378519.

- 23. MacMahon S, Sharpe N, Gamble G, Clague A, Mhurchu CN, Clark T, et al. Randomized, placebo-controlled trial of the angiotensin-converting enzyme inhibitor, ramipril, in patients with coronary or other occlusive arterial disease. PART-2 Collaborative Research Group. Prevention of Atherosclerosis with Ramipril. J Am Coll Cardiol. 2000; 36(2):438–43. Epub 2000/08/10. PMID: 10933355.
- 24. Pitt B, Byington RP, Furberg CD, Hunninghake DB, Mancini GB, Miller ME, et al. Effect of amlodipine on the progression of atherosclerosis and the occurrence of clinical events. PREVENT Investigators. Circulation. 2000; 102(13):1503–10. Epub 2000/09/27. PMID: 11004140.
- 25. Randomised trial of a perindopril-based blood-pressure-lowering regimen among 6,105 individuals with previous stroke or transient ischaemic attack. Lancet. 2001; 358(9287):1033–41. Epub 2001/10/09. https://doi.org/10.1016/S0140-6736(01)06178-5 PMID: 11589932.
- 26. Patel A, MacMahon S, Chalmers J, Neal B, Woodward M, Billot L, et al. Effects of a fixed combination of perindopril and indapamide on macrovascular and microvascular outcomes in patients with type 2 diabetes mellitus (the ADVANCE trial): a randomised controlled trial. Lancet. 2007; 370(9590):829–40. Epub 2007/09/04. https://doi.org/10.1016/S0140-6736(07)61303-8 PMID: 17765963.
- Teo KK, Burton JR, Buller CE, Plante S, Catellier D, Tymchak W, et al. Long-term effects of cholesterol lowering and angiotensin-converting enzyme inhibition on coronary atherosclerosis: The Simvastatin/ Enalapril Coronary Atherosclerosis Trial (SCAT). Circulation. 2000; 102(15):1748–54. Epub 2000/10/ 12. PMID: 11023927.
- Staessen JA, Fagard R, Thijs L, Celis H, Arabidze GG, Birkenhager WH, et al. Randomised doubleblind comparison of placebo and active treatment for older patients with isolated systolic hypertension. The Systolic Hypertension in Europe (Syst-Eur) Trial Investigators. Lancet. 1997; 350(9080):757–64. Epub 1997/09/23. PMID: 9297994.
- Anderson KM, Odell PM, Wilson PW, Kannel WB. Cardiovascular disease risk profiles. Am Heart J. 1991; 121(1 Pt 2):293–8. Epub 1991/01/01. PMID: 1985385.
- Gaziano TA, Steyn K, Cohen DJ, Weinstein MC, Opie LH. Cost-effectiveness analysis of hypertension guidelines in South Africa: absolute risk versus blood pressure level. Circulation. 2005; 112(23):3569– 76. Epub 2005/12/07. https://doi.org/10.1161/CIRCULATIONAHA.105.535922 PMID: 16330698.
- **31.** Montgomery AA, Fahey T, Ben-Shlomo Y, Harding J. The influence of absolute cardiovascular risk, patient utilities, and costs on the decision to treat hypertension: a Markov decision analysis. J Hypertens. 2003; 21(9):1753–9. Epub 2003/08/19. <u>https://doi.org/10.1097/01.hjh.0000084708.87421.d1</u> PMID: 12923409.
- Wright JT Jr., Williamson JD, Whelton PK, Snyder JK, Sink KM, Rocco MV, et al. A Randomized Trial of Intensive versus Standard Blood-Pressure Control. N Engl J Med. 2015; 373(22):2103–16. Epub 2015/ 11/10. https://doi.org/10.1056/NEJMoa1511939 PMID: 26551272; PubMed Central PMCID: PMCPMC4689591.
- Lonn EM, Bosch J, Lopez-Jaramillo P, Zhu J, Liu L, Pais P, et al. Blood-Pressure Lowering in Intermediate-Risk Persons without Cardiovascular Disease. N Engl J Med. 2016; 374(21):2009–20. Epub 2016/ 04/05. https://doi.org/10.1056/NEJMoa1600175 PMID: 27041480.
- Dahabreh IJ, Kent DM. Index event bias as an explanation for the paradoxes of recurrence risk research. JAMA. 2011; 305(8):822–3. Epub 2011/02/24. https://doi.org/10.1001/jama.2011.163 PMID: 21343582; PubMed Central PMCID: PMCPmc4115579.
- Clayton TC, Lubsen J, Pocock SJ, Voko Z, Kirwan BA, Fox KA, et al. Risk score for predicting death, myocardial infarction, and stroke in patients with stable angina, based on a large randomised trial cohort of patients. BMJ. 2005; 331(7521):869. Epub 2005/10/08. https://doi.org/10.1136/bmj.38603.656076.
 PMID: 16210253; PubMed Central PMCID: PMCPmc1255789.
- 36. Dorresteijn JA, Visseren FL, Wassink AM, Gondrie MJ, Steyerberg EW, Ridker PM, et al. Development and validation of a prediction rule for recurrent vascular events based on a cohort study of patients with arterial disease: the SMART risk score. Heart. 2013; 99(12):866–72. Epub 2013/04/12. https://doi.org/ 10.1136/heartjnl-2013-303640 PMID: 23574971.
- Bhatt DL, Eagle KA, Ohman EM, Hirsch AT, Goto S, Mahoney EM, et al. Comparative determinants of 4-year cardiovascular event rates in stable outpatients at risk of or with atherothrombosis. JAMA. 2010; 304(12):1350–7. Epub 2010/09/02. https://doi.org/10.1001/jama.2010.1322 PMID: 20805624.
- Chobanian AV, Bakris GL, Black HR, Cushman WC, Green LA, Izzo JL Jr., et al. The Seventh Report of the Joint National Committee on Prevention, Detection, Evaluation, and Treatment of High Blood Pressure: the JNC 7 report. JAMA. 2003; 289(19):2560–72. Epub 2003/05/16. https://doi.org/10.1001/jama. 289.19.2560 PMID: 12748199.
- 39. Whelton PK, Carey RM, Aronow WS, Casey DE Jr., Collins KJ, Dennison Himmelfarb C, et al. 2017 ACC/AHA/AAPA/ABC/ACPM/AGS/APhA/ASH/ASPC/NMA/PCNA Guideline for the Prevention, Detection, Evaluation, and Management of High Blood Pressure in Adults: A Report of the American College of Cardiology/American Heart Association Task Force on Clinical Practice Guidelines. Hypertension.

2017;Epub ahead of print. Epub 2017/11/15. https://doi.org/10.1161/hyp.00000000000065 PMID: 29133356.

40. Allen N, Berry JD, Ning H, Van Horn L, Dyer A, Lloyd-Jones DM. Impact of blood pressure and blood pressure change during middle age on the remaining lifetime risk for cardiovascular disease: the cardiovascular lifetime risk pooling project. Circulation. 2012; 125(1):37–44. Epub 2011/12/21. https://doi.org/ 10.1161/CIRCULATIONAHA.110.002774 PMID: 22184621; PubMed Central PMCID: PMCPmc3310202.